Author's response to reviews

Title: Online follow-up of individuals with gastroesophageal reflux disease using a patient-reported outcomes instrument: results of an observational study

Authors:

Merel M Tielemans (M.Tielemans@MDL.umcn.nl)
Martijn GH van Oijen (MVanOijen@mednet.ucla.edu)

Version: 2 Date: 26 April 2013

Author's response to reviews:

I have added the following ethics statement in the Method Section of the manuscript:

The Medical Ethical Committee of the Radboud University Nijmegen assessed the research proposal of this study and concluded that it could be waived for ethical review, as we did not gather information about participants from other sources (e.g. medical records) and data storage occurred in accordance with Dutch law. For this reason, we did not obtain written informed consent of all participants.